Author’s response to reviews

Title: Randomized clinical trial of efficacy and safety of colchicine 0.5% cream versus photodynamic therapy with methyl aminolevulinate in treatment of skin field cancerization: study protocol

Authors:
Anna Miola (anna_fmrp@yahoo.com.br)
Eliane Roio (eliane_roio@yahoo.com.br)
Helio Miot (heliomiot@gmail.com)
Luciana Abbade (lfabbade@fmb.unesp.br)
Juliano Schimitt (julivs@gmail.com)

Version: 1 Date: 10 Oct 2016

Author’s response to reviews:

Dear Dafne Solera

Thank you for commenting this study protocol and help to improve it. We need to apologize about the late.

The points raised in decision letter will be answered in the order below, according to the email.

1. Ethical and Funding Approval Documentation

The ethical aproval as sent to email BMCSeriesEditorial@biomedcentral.com.

2. Funding
This study received no specific grant from any funding agency in the public, commercial or not-for-profit sectors.

3. Study status

This study is ongoing. We are collecting patient data and introducing interventions. Therefore, we do not have results after the implementation of the measures.

4. Related Articles

There is no any publications containing this study protocol or the results of this study, because it is ongoing at the moment.

Thank you for decision letter and we are available for any clarification.

Best wishes,

Anna Carolina Miola